Author's response to reviews

Title: In favour of the definition "adolescents with idiopathic scoliosis": juvenile and adolescent idiopathic scoliosis braced after ten years of age, do not show different end results. SOSORT Award Winner 2014.

Authors:

Fabio Zaina (fabio.zaina@isico.it)
Monia Lusini (monia.lusini@isico.it)
Salvatore Minnella (salvatore.minnela@isico.it)
Stefano Negrini (stefano.negrini@isico.it)

Version: 2 Date: 26 May 2014

Author's response to reviews:

Literature concerning the topic of Juvenile idiopathic scoliosis is poor. The comparisons proposed in the present study, are very original and offer some very interesting issues in the clinical approach of juvenile idiopathic scoliosis, so the clinical relevance is evident. This paper was chosen by the scientific board of SOSORT as winner of the AWARD 2014.